

eCommons@AKU

Department of Paediatrics and Child Health

Division of Woman and Child Health

10-15-2020

Response of extensively drug resistant salmonella typhi to treatment with meropenem and azithromycin, in Pakistan

Sonia Qureshi

Abdullah B. Naveed

Mohammad Tahir Yousafzai

Khalil Ahmad

Sarwat Ansari

See next page for additional authors

Follow this and additional works at: https://ecommons.aku.edu/ pakistan_fhs_mc_women_childhealth_paediatr

Part of the Bacteria Commons, Bacterial Infections and Mycoses Commons, Pediatrics Commons, and the Women's Health Commons

Authors

Sonia Qureshi, Abdullah B. Naveed, Mohammad Tahir Yousafzai, Khalil Ahmad, Sarwat Ansari, Heeramani Lohana, Aiman Mukhtar, and Farah Naz Qamar

RESEARCH ARTICLE

Response of extensively drug resistant *Salmonella* Typhi to treatment with meropenem and azithromycin, in Pakistan

Sonia Qureshi¹, Abdullah B. Naveed², Mohammad Tahir Yousafzai¹, Khalil Ahmad¹, Sarwat Ansari³, Heeramani Lohana³, Aimen Mukhtar¹, Farah Naz Qamar¹*

1 Department of Pediatrics and Child Health, Aga Khan University Hospital, Karachi, Pakistan, 2 Medical College, Aga Khan University, Karachi, Pakistan, 3 Department of Pediatric and Child Health, Aga Khan Maternal and Child Care Centre, Hyderabad, Pakistan

* farah.qamar@aku.edu

Abstract

Introduction

Salmonella Typhi is one of the leading health problems in Pakistan. With the emergence of extensively drug resistant (XDR) *Salmonella* Typhi, treatment options are limited. Here we report the clinical manifestations and the response to treatment of patients with XDR Typhoid fever. The patients were treated with either Meropenem or Azithromycin or a combination of both.

Methods

We reviewed the records of culture confirmed XDR typhoid who visited Aga Khan University Hospital (AKUH), Karachi and Aga Khan Secondary Care Hospital, Hyderabad from April 2017 to June 2018. Symptoms developed during disease, unplanned treatment extension and complications developed while on antimicrobials was recorded. Means with standard deviation were calculated for duration of treatment, time to defervescence, and cost of treatment.

Results

Records of 81 culture confirmed XDR typhoid patients admitted at the AKU hospitals were reviewed. Most, (n = 45; 56%) were male. Mean age of the cases was 8.03 years with range (1–40). About three quarter (n = 66) of the patients were treated as inpatient. Fever and vomiting were the most common symptoms at the time of presentation. Oral azithromycin alone (n = 22; 27%), intravenous meropenem alone (n = 20; 25%), or a combination of azi-thromycin and meropenem (n = 39; 48%) were the options used for treatment. Average (95% confidence interval) time to defervescence was 7.1(5.5–8.6), 6.7(4.7–8.7), and 6.7 (5.5–7.9) days for each treatment option respectively whereas there were 1,0 and 3 treatment failures in each treatment option respectively. Average cost of treatment per day for azithromycin was US\$5.87 whereas it was US\$88.46 for meropenem.



G OPEN ACCESS

Citation: Qureshi S, Naveed AB, Yousafzai MT, Ahmad K, Ansari S, Lohana H, et al. (2020) Response of extensively drug resistant *Salmonella* Typhi to treatment with meropenem and azithromycin, in Pakistan. PLoS Negl Trop Dis 14(10): e0008682. https://doi.org/10.1371/journal. pntd.0008682

Editor: Jessica N. Ricaldi, Centers for Disease Control and Prevention, UNITED STATES

Received: July 7, 2019

Accepted: August 5, 2020

Published: October 15, 2020

Copyright: © 2020 Qureshi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The authors received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

Patients treated with either Azithromycin, Meropenem alone or in combination showed similar time to defervescence. Because of the lower cost of azithromycin, it is preferable in lower socio-economic areas. Background estimates for power calculation can be made for more robust clinical trials using this observational data.

Author summary

Ever since the emergence of Extensively Drug Resistant (XDR) Typhoid fever in Pakistan, which is resistant to all but two classes of anti-microbials, we need to find an effective treatment plan so that *S*.Typhi does not also make these anti-microbials redundant in the future. In this retrospective case series review, we observed the response of XDR Typhoid fever patients treated with meropenem or azithromycin or combination of both. All 3 groups showed a similar therapeutic response based on number of complications developed and time taken for fever clearance. The higher price of meropenem however could become a deterrent to its use in poor socio-economic areas. This new knowledge may contribute to better treatment plans in handling XDR enteric fever in the future whilst also serving as a rough guide in how to treat patients with XDR Typhoid fever in the present.

Introduction

Typhoid fever is a systemic infection caused by the *Salmonella* enterica serovar Typhi (*S*. Typhi). According to the World Health Organisation (WHO), estimates of burden of typhoid fever range between 11 and 21 million cases and roughly 128 000 to 161 000 deaths annually [1]. Typhoid fever is a disease of low and middle-income countries but with the emergence of antimicrobial resistance and increase in international travel, the threat is now global[2]. In the last two decades several outbreaks of multi-drug resistant (MDR) *S*.Typhi of H58 genotype have been reported from various parts of the world[3,4]. These organisms are a major threat to typhoid treatment as they are resistant to the first line antimicrobials; ampicillin, chloramphenicol, and trimethoprim-sulfamethoxazole. Similarly, resistance to fluoroquinolones has also increased over the same period. Consequently, third-generation cephalosporins, particularly ceftriaxone, have become the antimicrobials of choice for typhoid treatment in endemic countries [5,6].

In November 2016, a large outbreak of ceftriaxone-resistant typhoid fever started in Hyderabad city of southern Pakistan. The associated organism was a H58 S.Typhi exhibiting resistance to five classes of antimicrobials (chloramphenicol, ampicillin, trimethoprim-sulfamethoxazole, fluoroquinolones, and third-generation cephalosporins) and was consequently labeled as extensively drug-resistant (XDR) S.Typhi[7,8]. This strain of XDR S. Typhi was sensitive to azithromycin and meropenem. The outbreak first started in Hyderabad city with it quickly spreading to the adjacent cities including Karachi. As of August, 2019, more than 10,000 cases of XDR Typhoid have been reported from Hyderabad and Karachi alone[9].

Owing to the novelty of XDR S.Typhi, no treatment guidelines are available yet globally. Treatment options are limited and no data on treatment outcomes of XDR Typhoid is available. The data we have is much needed for the development of treatment protocols for XDR Typhoid patients. Moreover, observational data from this study may also be helpful in designing a more robust clinical trial for the treatment of XDR Typhoid in the future.

Methods

Study setting and population

A retrospective review of XDR Typhoid cases was conducted in two hospitals; Aga Khan University Hospital (AKUH) in Karachi and Aga Khan Maternal and Child Centre (AKMCC) in Hyderabad. AKUH is a 740-bed private tertiary care teaching hospital located in the metropolitan city of Karachi while AKMCC is a private secondary care hospital catering only to mothers and children, located in the city of Hyderabad. Karachi and Hyderabad are located about 150 kilometers away from each other in the province of Sindh, Pakistan.

Patients with a diagnosis of blood culture confirmed XDR S.Typhi were identified from the registry of both hospitals. Hospital records of typhoid patients treated as either inpatient or outpatient during 1st April 2017 to 30th June 2018 were reviewed for screening to be included in this study. Typhoid patients with blood culture confirmation for S.Typhi and resistance to the five classes of antibiotics (ampicillin, chloramphenicol, trimethoprim-sulfamethoxazole, fluoroquinolone, and 3rd generation cephalosporin (ceftriaxone or cefixime) were eligible to be included in this study. Patients with incomplete medical records especially missing information on antimicrobial treatment, duration of treatment, treatment failure, and time to defervescence were excluded from the study. Also, patients who left against medical advice (LAMA) within 2 days of admission, those who had a positive blood culture but did not seek treatment at either hospital or those who did not return for follow-up visit in clinic after the confirmation of blood culture testing were excluded. A total of 142 records with the diagnosis of typhoid were screened and only 81 records which fulfilled the eligibility criteria were included in this study (Fig 1).

Data collection tool and operational definitions

We used a structured tool in English language to extract information from the medical records. The tool was composed of variables such as demographic characteristics (age, gender), symptoms including fever at the time of presentation to the hospital/clinic, blood culture and antimicrobial sensitivity findings, time in days to defervescence, complications (if any) of disease process, type and duration of antimicrobial therapy, side effects (if any) and treatment





https://doi.org/10.1371/journal.pntd.0008682.g001

failure. Clinical response of antimicrobial therapy was defined as time taken for defervescence (time from first recording of fever, either inpatient or outpatient, till return of body temperature to less than 38[°] C for more than 48 hours). Treatment failure was defined as either relapse of disease within 30 days of treatment completion, or development of complications while on therapy (septic shock, meningitis, death, hepatitis etc.[10]), or extension of duration of antimicrobial therapy beyond 14 days for meropenem and 10 days for azithromycin. Cost of purchase of anti-microbials was calculated for maximum dosages that could theoretically be given. This meant that meropenem cost was calculated for 2g every 8 hours while azithromycin cost was calculated for 1g every 24 hours. No other direct or indirect costs were measured.

Ethical considerations

This study was conducted as a part of a larger study which involved investigation of the outbreak of XDR Typhoid and control interventions in Hyderabad and Karachi. The larger study was approved by ethical review committee (ERC) of Aga Khan University Karachi, Pakistan and national bioethics committee (NBC) of Pakistan. For this specific part of the study which involved retrospective review of hospital records, we submitted it for an ERC exemption which was granted.

Data management and analysis

Two research assistants entered the data independently using the statistical package for social scientists (SPSS) version 25.0. The two datasets were compared to assess for any typographic/ data entry errors, validated against the corresponding hard copy of the data tools and corrected. Data was also checked for missing information on the essential variables such as antimicrobial treatment, duration of treatment and time to defervescence. Any variable with extreme values or illogical values were validated against the corresponding physical data tool and the actual medical record files. Frequencies with percentages were calculated for categorical variables such as gender, type of treatment, place (hospital) of treatment, and inpatient versus outpatient treatment. In addition, means with standard deviation were calculated for continuous variables such as age in years, duration of treatment, and time to defervescence etc. Cost of treatment per day for each type of antimicrobial therapy was also calculated and compared across the groups.

Results

Out of 81 records reviewed for this study, 47(58%) were from AKMCC. About three quarter (n = 66; 81%) of XDR Typhoid patients were treated as inpatient. Slightly more than half (n = 45; 56%) of the patients were male and average age of the patients was 8.0 years with range (1–40) (Fig 2). Average temperature at the time of presentation in the hospital was 38.5 \pm 0.89 (Range = 40.0–36.0) ^oC. 64% (n = 52) of patients in this study were started on empirical anti-microbial therapy prior to presentation; either self-prescribed or on the recommendation of a physician (Table 1).

Fever (n = 81; 100%) and vomiting (n = 53; 65%) were the most common symptoms developed during course of disease, followed by diarrhea (47%), abdominal pain (43%) and head-ache (38%) (Fig 3).

Out of 81 XDR Typhoid patients, 22 (27%) were treated with azithromycin alone, 20 (25%) with meropenem alone and 39 (48%) received a combination of azithromycin and meropenem. Patients who received meropenem only were all treated as inpatient, whereas only 11/22 (50%) patients treated with azithromycin required hospitalization. Meropenem was given Intravenously (IV) at a dose of 20-40mg/kg three times a day while azithromycin was given





https://doi.org/10.1371/journal.pntd.0008682.g002

orally at 20mg/kg/day. There was only one treatment failure among patients treated with azithromycin while on the other hand three treatment failures each were observed among the patients treated with combination of azithromycin and meropenem. No treatment failure was seen in patients treated with meropenem. Average time in days (95% confidence interval) to defervescence was 7.1(5.5–8.6) for azithromycin, 6.7(4.7–8.7) for meropenem, and 6.7(5.5– 7.9) for combination of meropenem and azithromycin. Average time to defervescence was lower for the patients treated with meropenem and combination of azithromycin and meropenem as compared to the patients treated with azithromycin (Fig 4). Average daily cost of treatment was highest (US\$94.33) for the patients receiving combination of azithromycin and meropenem and lowest (US\$5.87) for those treated with azithromycin alone (Table 2).

Discussion

In this study we assessed the response of azithromycin and meropenem–either alone or in combination, for the treatment of XDR Typhoid fever. All three treatment options were highly effective, with less than 5% treatment failure in each group, however patients treated with meropenem had no treatment failure. Patients treated with meropenem and combination of azi-thromycin and meropenem had a slightly earlier time to deferve cence.

Patient characteristics	Frequency (%)	
Centers*		
АКИН	34 (42.0%)	
AKMCCC	47 (58.0%)	
Age in years at diagnosis (Mean)	8.03	
(Min-Max)	1-40	
Gender (Male)	45 (56.0%)	
Patients treated as Inpatients	66 (81.0%)	
Fever (°C) at time of hospital admission (Mean \pm SD)	38.5 ± 0.89	
Use of Antibiotics in past 2 weeks	52 (64.2%)	
Use of Anti-Pyretics in past 2 weeks	54 (66.6%)	
Use of Anti-Diarrheal in past 2 weeks	3 (3.7%)	

Table 1. Demographic characteristics and treatment of culture proven XDR Typhoid patients (N = 81).

*AKUH = Aga Khan University Hospital, AKMCCC = Aga Khan Maternal and Child Care Centre Hyderabad.

https://doi.org/10.1371/journal.pntd.0008682.t001



Fig 3. Presenting symptoms of patients at time of hospital encounter arranged in descending order of occurrence. https://doi.org/10.1371/journal.pntd.0008682.g003

No evidence is available to establish fever clearance time for XDR typhoid. Data from previous studies for MDR typhoid and fluoroquinolone resistant strains, has shown that defervescence is achieved in less than 7 days [11-14]. These results are in line with the expected time to defervescence for non-resistant typhoid fever which is 3.5-8 days from initiation of antimicrobial therapy[15,16]. Our study was consistent with this range and showed all 3 treatment arm groups achieving defervescence within the given range. Meropenem is usually considered as the last option for multi drug resistant gram-negative infections[17] hence there are limited reports that establish a definitive time to fever clearance for typhoid fever when treated with meropenem. Literature search reveals 5 fluoroquinolone resistant S.Typhi strains that were treated with meropenem alone [18,19] or as part of combination therapy [20-22]. Out of these, only two of the cases achieved defervescence within the expected range [18,19] when treated with meropenem alone while the remaining three cases required augmentation of meropenem with another antimicrobial to achieve defervescence [20-22]. More robust data is available for azithromycin-based therapies where multiple clinical trials have shown that defervescence is achieved within the expected range [23]. A study by Nair et al suggested that fever clearance time was longer when patients were treated with azithromycin compared to when patients



Fig 4. Comparison of time to defervescence across different types of antimicrobial therapy for treating XDR Typhoid patients.

https://doi.org/10.1371/journal.pntd.0008682.g004

Anti-microbial Group	N (%)	Treated as Inpatients n (%)	Treatment Failure n (%)	Duration of Treatment (days) (Mean ± SD)	Cost of Treatment per day ‡
Meropenem Only	20 (24.7%)	20 (100%)	0 (0%)	8.1 ± 2.5	13800PKR (US\$88.46)
Azithromycin Only	22 (27.2%)	11 (50%)	1 (4.5%)	6.6 ± 2.7	915PKR (US\$5.87)
Combination of Meropenem and Azithromycin	39 (48.1%)	35 (89.7%)	3 (4.5%)	Azithromycin: 7.5 ± 3.8 Meropenem: 8.5 ± 4.3	14715PKR (US\$94.33)

Table 2. Clinical Outcome of patients among the three treatment groups.

‡ All prices given in local currency and converted to international dollar according to Bureau of the Fiscal Services Treasury Reporting Rates of Exchange

https://doi.org/10.1371/journal.pntd.0008682.t002

were treated with ceftriaxone[24]. However, in our study, fever clearance time for XDR typhoid fever patients treated with either azithromycin or meropenem or a combination of meropenem and azithromycin was similar.

The treatment failure that occurred in azithromycin only group was because of non-resolving fever even after being given recommended dose and duration of antimicrobial. The fever resolved after extension of antimicrobial therapy by 5 days. No underlying condition (immune suppression) or disease complication or non-compliance to therapy was recorded. The treatment failure percentage with azithromycin in our study is consistent with previous clinical trials that have established the efficacy of azithromycin in treating enteric fever[23]. Frenck et al [25] and Girgis et al [26] explained that the intracellular accumulation of azithromycin within cells, its secretion into the biliary tree and the long half-life of the drug make it extremely effective against intracellular infections like enteric fever. Literature, however, is inconclusive when it comes to the efficacy of meropenem for treating enteric fever. Four case reports point to its clinical inefficacy based on either relapse of disease[19] or augmentation of meropenem with another antimicrobial due to unsatisfactory clinical results with meropenem alone [20-22]. Godbole et al^[20] discussed the possible treatment failures due to the limited intracellular penetration of meropenem whereas Blumentrath et al^[19] explained the treatment failures via the phenomena of tolerance and persistence exhibited by S.Typhi. In contrast, Munir et al^[18] showed favorable results when treating XDR Typhoid fever with meropenem. Our study patients showed adequate results when treated with meropenem. Meropenem is a type of β lactam antibiotic with inherent stability against ESBLs and possesses a broad-spectrum activity, these factors could explain the clearance percentage for meropenem[27]. In the group treated with combination of meropenem and azithromycin, there were 3 treatment failures. All 3 patients developed systemic complications and unfortunately, one of the patients passed away. This could be attributed to delay in treatment of both patients because of delayed presentation to the hospital resulting in an unfavorable clinical picture on arrival. The preference (n = 39; 48%) for using both antibiotics together can likely be explained by the fact that one is a bacteriostatic (meropenem) and the other is a bactericidal (azithromycin). Past studies have shown that the combination of ceftriaxone (a bactericidal) with azithromycin (a bacteriostatic) reduced fever clearance times [28], albeit no study could be found assessing the synergistic action of meropenem with azithromycin for treating Enteric fever. Case reports by Kleine et al [21], Godbole et al^[20] and Blumentrath et al^[19] also suggested that meropenem be combined with another antimicrobial agent for treating fluoroquinolone resistant and XDR typhoid fever. Since there are no evidence based clinical guidelines for the treatment of XDR Typhoid, azithromycin is a good choice since it accumulates intracellularly and has a long half-life.

The longer duration of treatment with azithromycin (6.6 ± 2.7 days) in our study compared to established protocols[23] was likely due to treating physicians fearing a relapse of the disease

and hence giving extended doses of azithromycin. The 1 treatment failure which required extension of antimicrobial therapy in this group also slightly skewed the mean. For meropenem however, there are only limited case reports that guide us about the dosing and duration of meropenem for XDR typhoid fever or fluoroquinolone resistant typhoid fever. Blumentrath et al[19] administered it for 10 days where as Kleine et al[21] had to administer it for 22 days which shows the variance in time meropenem takes to achieve fever clearance. The duration of treatment with meropenem in our study was similar to duration of treatment of Typhoid fever with ceftriaxone in past clinical trials[29]. We believe this was due to the similar mechanism of action of the two antimicrobials (both are bactericidal) and short half-life's (60mins for meropenem vs 330mins for ceftriaxone)[30][31].

Another facet to consider from this study is the cost of therapy of meropenem compared with azithromycin. Enteric fever is endemic in countries with a lower socio-economic status due to difficulty in maintaining hygienic conditions. Patients on meropenem therapy must incur an average daily cost of US\$88.46 compared to US\$5.87 for azithromycin. For patients with a lower socio-economic status, the extra cost can prove to be a deterrent in obtaining the optimum therapy.

Since this is a retrospective study, the information on patient follow ups is limited and the detailed cost of hospitalization was not recorded. The criteria for assessing which patient necessitated a specific therapy was also subjective and based on the discretion of the treating physician Also, we believe the study might have had selection bias because of patients who were lost to follow up and those who were excluded based on lack of proper data. Furthermore, as there is an uneven receipt of antibiotics received before presentation to hospital, the average time to defervescence could have been affected across groups, though none of them had either received azithromycin or meropenem. The strength of this study was the inclusion of cases based only on positive blood cultures and not clinical judgement.

In conclusion either azithromycin or meropenem alone or in combination were used for the treatment of XDR Typhoid in an outbreak setting in Pakistan. The time to defervescence and the percentage of treatment failures in all groups were similar. Azithromycin is a convenient choice based on cost of therapy and availability in oral form. Clinical trials are needed to establish days of treatment required and the best treatment options for XDR Typhoid fever.

Supporting information

S1 Supporting Information. STROBE checklist. (DOCX)

Acknowledgments

We thank the children and families of Hyderabad and Karachi for agreeing to participate in this study.

Author Contributions

Conceptualization: Sonia Qureshi, Mohammad Tahir Yousafzai, Farah Naz Qamar.

Data curation: Sonia Qureshi, Abdullah B. Naveed, Mohammad Tahir Yousafzai, Sarwat Ansari, Heeramani Lohana, Aimen Mukhtar.

Formal analysis: Mohammad Tahir Yousafzai.

Investigation: Abdullah B. Naveed, Farah Naz Qamar.

Methodology: Abdullah B. Naveed, Aimen Mukhtar.

Project administration: Sonia Qureshi, Mohammad Tahir Yousafzai.

Software: Abdullah B. Naveed.

Supervision: Sonia Qureshi, Khalil Ahmad, Farah Naz Qamar.

Validation: Sonia Qureshi, Farah Naz Qamar.

Writing – original draft: Abdullah B. Naveed.

Writing – review & editing: Sonia Qureshi, Mohammad Tahir Yousafzai, Khalil Ahmad, Farah Naz Qamar.

References

- World Health Organisation. Typhoid vaccine: WHO position paper—March 2018. Wkly Epidemiol Rec. 2018.
- Saleem Z, Hassali MA. Travellers take heed: Outbreak of extensively drug resistant (XDR) typhoid fever in Pakistan and a warning from the US CDC. Travel Medicine and Infectious Disease. 2019. https://doi.org/10.1016/j.tmaid.2018.10.013 PMID: 30339826
- Muti M, Gombe N, Tshimanga M, Takundwa L, Bangure D, Mungofa S, et al. Typhoid outbreak investigation in Dzivaresekwa, suburb of Harare City, Zimbabwe, 2011. Pan Afr Med J. 2014. <u>https://doi.org/ 10.11604/pamj.2014.18.309.4288 PMID: 25469202</u>
- Lewis MD, Serichantalergs O, Pitarangsi C, Chuanak N, Mason CJ, Regmi LR, et al. Typhoid Fever: A Massive, Single-Point Source, Multidrug-Resistant Outbreak in Nepal. Clin Infect Dis. 2005. <u>https://doi.org/10.1086/427503</u> PMID: 15712078
- Andrews JR, Qamar FN, Charles RC, Ryan ET. Extensively Drug-Resistant Typhoid—Are Conjugate Vaccines Arriving Just in Time? N Engl J Med. 2018. https://doi.org/10.1056/nejmp1803926 PMID: 30332569
- Qamar FN, Yousafzai MT, Sultana S, Baig A, Shakoor S, Hirani F, et al. A Retrospective Study of Laboratory-Based Enteric Fever Surveillance, Pakistan, 2012–2014. J Infect Dis. 2018. <u>https://doi.org/10.1093/infdis/jiy205</u> PMID: 30060168
- Klemm EJ, Shakoor S, Page AJ, Qamar FN, Judge K, Saeed DK, et al. Emergence of an Extensively Drug-Resistant Salmonella enterica Serovar Typhi Clone Harboring a Promiscuous Plasmid Encoding Resistance to Fluoroquinolones and Third-Generation Cephalosporins. Klugman Neil Wren, Brendan Levine, KPF, editor. MBio. 2018; 9: e00105 https://doi.org/10.1128/mBio.00105-18 PMID: 29463654
- Qamar FN, Yousafzai MT, Khalid M, Kazi AM, Lohana H, Karim S, et al. Outbreak investigation of ceftriaxone-resistant Salmonella enterica serotype Typhi and its risk factors among the general population in Hyderabad, Pakistan: a matched case-control study. Lancet Infect Dis. 2018. https://doi.org/10.1016/ S1473-3099(18)30483-3
- WHO. No Title. Wkly Epidemiol Rec. 2019;12. Available: <u>https://reliefweb.int/sites/reliefweb.int/files/</u> resources/Epi_Monitor_2019_12_34.pdf
- Malik AS. Complications of bacteriologically confirmed typhoid fever in children. J Trop Pediatr. 2002. https://doi.org/10.1093/tropej/48.2.102 PMID: 12022423
- Chinh NT, Parry CM, Ly NT, Ha HD, Thong MX, Diep TS, et al. A Randomized Controlled Comparison of Azithromycin and Ofloxacin for Treatment of Multidrug-Resistant or Nalidixic Acid-Resistant Enteric Fever. Antimicrob Agents Chemother. 2000; 44: 1855 LP- 1859. <u>https://doi.org/10.1128/AAC.44.7.</u> 1855–1859.2000
- Bandyopadhyay R, Balaji V, Yadav B, Jasmine S, Sathyendra S, Rupali P. Effectiveness of treatment regimens for Typhoid fever in the nalidixic acid-resistant S. typhi (NARST) era in South India. Trop Doct. 2018; 48: 182–188. https://doi.org/10.1177/0049475518758884 PMID: 29495943
- Kumar R, Gupta N, Shalini. Multidrug—Resistant typhoid fever. Indian J Pediatr. 2007. <u>https://doi.org/10.1007/s12098-007-0024-z PMID: 17264451</u>
- Wallace MR, Yousif AA, Mahroos GA, Mapes T, Threlfall EJ, Rowe B, et al. Ciprofloxacin versus ceftriaxone in the treatment of multiresistant typhoid fever. Eur J Clin Microbiol Infect Dis. 1993; 12: 907– 910. https://doi.org/10.1007/BF01992163 PMID: 8187784

- Lasserre R, Sangalang RP, Santiago L. Three-day treatment of typhoid fever with two different doses of ceftriaxone, compared to 14-day therapy with chloramphenicol: a randomized trial. J Antimicrob Chemother. 1991; 28: 765–772. https://doi.org/10.1093/jac/28.5.765 PMID: 1778879
- 16. Ray P, Sharma J, Marak RSK, Garg RK. Predictive efficacy of nalidixic acid resistance as a marker of fluoroquinolone resistance in Salmonella enterica var Typhi. Indian J Med Res. 2006.
- Exner M, Bhattacharya S, Christiansen B, Gebel J, Goroncy-Bermes P, Hartemann P, et al. Antibiotic resistance: What is so special about multidrug-resistant Gram-negative bacteria? GMS Hyg Infect Control. 2017; 12: Doc05. https://doi.org/10.3205/dgkh000290 PMID: 28451516
- Munir T, Lodhi M, Ansari JK, Andleeb S, Ahmed M. Extended Spectrum Beta Lactamase producing Cephalosporin resistant Salmonella Typhi, reported from Rawalpindi, Pakistan. J Pak Med Assoc. 2016; 66: 1035–1036. PMID: 27524545
- Blumentrath CG, Müller G, Teichmann D, Tiesmeier J, Petridou J. Relapse of typhoid fever following delayed response to meropenem: A case report and review of previously published cases indicating limited clinical efficacy of meropenem for the treatment of typhoid fever. GMS Ger Med Sci. 2019. <u>https://</u> doi.org/10.3205/000267 PMID: 30837820
- Godbole GS, Day MR, Murthy S, Chattaway MA, Nair S. First Report of CTX-M-15 Salmonella Typhi From England. Clin Infect Dis. 2018; 66: 1976–1977. <u>https://doi.org/10.1093/cid/ciy032</u> PMID: 29471386
- Kleine C-E, Schlabe S, Hischebeth GTR, Molitor E, Pfeifer Y, Wasmuth J-C, et al. Successful Therapy of a Multidrug-Resistant Extended-Spectrum beta-Lactamase-Producing and Fluoroquinolone-Resistant Salmonella enterica Subspecies enterica Serovar Typhi Infection Using Combination Therapy of Meropenem and Fosfomycin. Clin Infect Dis. 2017; 65: 1754–1756. <u>https://doi.org/10.1093/cid/cix652</u> PMID: 29020162
- Lukácová L, Orságová I, Zjevíková A, Chmelarová E. Treatment failure in case of typhoid fever imported from India to Czech Republic, December 2008—January 2009. Euro Surveill. 2009.
- Effa EE, Bukirwa H. Azithromycin for treating uncomplicated typhoid and paratyphoid fever (enteric fever). Cochrane Database of Systematic Reviews. 2008. <u>https://doi.org/10.1002/14651858</u>. CD006083.pub2 PMID: 18843701
- Nair B, Simalti A, Sharma S. Study comparing ceftriaxone with azithromycin for the treatment of uncomplicated typhoid fever in children of India. Ann Trop Med Public Heal. 2017; 10: 205–210. <u>https://doi.org/ 10.4103/1755-6783.205534</u>
- Frenck RWJ, Nakhla I, Sultan Y, Bassily SB, Girgis YF, David J, et al. Azithromycin versus ceftriaxone for the treatment of uncomplicated typhoid fever in children. Clin Infect Dis. 2000; 31: 1134–1138. https://doi.org/10.1086/317450 PMID: 11073741
- 26. Girgis NI, Butler T, Frenck RW, Sultan Y, Brown FM, Tribble D, et al. Azithromycin versus ciprofloxacin for treatment of uncomplicated typhoid fever in a randomized trial in Egypt that included patients with multidrug resistance. Antimicrob Agents Chemother. 1999; 43: 1441–1444. <u>https://doi.org/10.1128/AAC.43.6.1441 PMID: 10348767</u>
- 27. Sorbera LA, Del Fresno N, Castañer RM, Rabasseda X. Faropenem Daloxate. Drugs Future. 2002; 27: 223. https://doi.org/10.1358/dof.2002.027.03.659134
- Zmora N, Shrestha S, Neuberger A, Paran Y, Tamrakar R, Shrestha A, et al. Open label comparative trial of mono versus dual antibiotic therapy for Typhoid Fever in adults. PLoS Negl Trop Dis. 2018. https://doi.org/10.1371/journal.pntd.0006380 PMID: 29684022
- Butler T. Treatment of typhoid fever in the 21st century: Promises and shortcomings. Clinical Microbiology and Infection. 2011. https://doi.org/10.1111/j.1469-0691.2011.03552.x PMID: 21722249
- Mouton JW, van den Anker JN. Meropenem Clinical Pharmacokinetics. Clinical Pharmacokinetics. 1995. https://doi.org/10.2165/00003088-199528040-00002 PMID: 7648757
- Yuk JH, Nightingale CH, Quintiliani R. Clinical Pharmacokinetics of Ceftriaxone. Clinical Pharmacokinetics. 1989. https://doi.org/10.2165/00003088-198917040-00002 PMID: 2686882